Author’s response to reviews

Title: Undergraduate medical education in emergency medical care: A nationwide survey at German medical schools

Authors:

Stefan K Beckers (sbeckers@ukaachen.de)
Arnd Timmermann (atimmer@zari.de)
Michael Müller (mp-mueller@web.de)
Matthias Angstwurm (matthias.angstwurm@med.uni-muenchen.de)
Felix Walcher (walcher@trauma.uni-frankfurt.de)

Version: 6 Date: 24 April 2009

Author’s response to reviews: see over
Cover Letter to Manuscript ID 4421879192326577

Undergraduate medical education in emergency medical care: A nationwide survey at German medical schools

Stefan K. Beckers; Arnd Timmermann; Michael Müller; Matthias Angstwurm and Felix Walcher

Dear Editor,

We were glad to hear, that the revised version of our manuscript is accepted for publication in principle. Thank you very much again for the opportunity to make the suggested modifications.

Again we checked thoroughly the orthography of the whole manuscript and consulted a native “reader” in order to correct remaining language issues.

In general we shortened some phrases to provide more readability and to fulfil recommendations of reviewer 2.

In addition, we checked and - where necessary - we adjusted the manuscript according to the provided checklist on the BiomedCentral-website, e.g. creating figures and data files, additional files and style of the references.

According to the raised general and reviewers’ comments we provide the following list point-by-point for the different sections were applicable:

**Title Page:**
As suggested by the editorial office we removed the phrase “part of the work was presented at …” from the title page.

**Background:**
We renamed the introduction into background and shortened phrases in this section to provide more readability.

**Methods:**
We renamed this section into “methods” and renamed legends and titles of the figures as well.
Thank you very much again for your kind consideration in this matter.

Best regards

[Signature]

Stefan K. Beckers

Section Emergency Medical Care, Department of Anaesthesiology
and AIXTRA – Aix-la-Chapelle Centre for Interdisciplinary Training in Medical Education
Medical Faculty RWTH Aachen University,
Pauwelsstr. 30, D – 52074 Aachen, Germany
e-mail: sbeckers@ukaachen.de
Phone: +49 (0) 241 – 80 89974
FAX: +49 (0) 241 – 80 82304